%%%%%%%%%%%%%%%%%%%% author.tex %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% sample root file for your "contribution" to a contributed volume
%
% Use this file as a template for your own input.
%
%%%%%%%%%%%%%%%% Springer %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


% RECOMMENDED %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\documentclass{svmult_jk}

% choose options for [] as required from the list
% in the Reference Guide

\usepackage{mathptmx}       % selects Times Roman as basic font
\usepackage{helvet}         % selects Helvetica as sans-serif font
\usepackage{courier}        % selects Courier as typewriter font
\usepackage{type1cm}        % activate if the above 3 fonts are
                            % not available on your system
%
\usepackage{makeidx}         % allows index generation
\usepackage{graphicx}        % standard LaTeX graphics tool
                             % when including figure files
\usepackage{multicol}        % used for the two-column index
\usepackage[bottom]{footmisc}% places footnotes at page bottom

% see the list of further useful packages
% in the Reference Guide

\makeindex             % used for the subject index
                       % please use the style svind.ist with
                       % your makeindex program

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}

\title*{How to Format an ARK-2010 Paper in LaTex}
% Use \titlerunning{Short Title} for an abbreviated version of
% your contribution title if the original one is too long
\author{J. Lenar\v ci\v c and P. Wenger}
% Use \authorrunning{Short Title} for an abbreviated version of
% your contribution title if the original one is too long
\institute{$^1$J. Stefan Institute, Ljubljana, Slovenia, \email{jadran.lenarcic@ijs.si} \\
$^2$Ecole Centrale de Nantes, France, \email{philippe.wenger@irccyn.ec-nantes.fr}}

%
%
% Use the package "url.sty" to avoid
% problems with special characters
% used in your e-mail or web address
%
\maketitle

\abstract{Accepted ARK articles will be published in an edited book
format by Springer. The format of this Example will result when using
the \emph{svmult} LaTeX class file. Note that this formatting (now
published by Springer) is similar but not identical to the one used in
the previous ARK editions which were published by Kluwer. Your
manuscript must be sent in Tex format with figures in Eps format to jadran.lenarcic@ijs.si. Please attach also a complete version in Pdf format for double-checking. The manuscript will be reviewed by two
independent reviewers.}

\keywords{Please indicate four to five keywords, which will also be used for composing the Subject Index.} %added JK

\section{Introduction}

The manuscript should be written in 8 pages. A manuscript of 6 or 10 pages may or may not be accepted, depending on the opinions of the reviewers and editors.

The paper should contain an abstract, an introduction, a main text, conclusions, and references.

The following are some examples of a typical equation, figure, table, citations and references. Please do not change the proposed formatting by introducing unappropriate commands in the text or by fixing the provided class file.


\subsection{Paragraphs and indents}

As you will see from the result of this sample paper, the class file will produce subsection titles that are all boldfaced. The indents of the paragraphs are done automatically. Please note that the first line of text that follows a heading is not indented, whereas the first lines of all subsequent paragraphs are. Instead of simply listing headings of different levels we recommend to let every heading be followed by at least a short passage of text.

For unnumbered lists we recommend to use the \emph{itemize} environment -- it will automatically render Springer's preferred layout.

\section{Equations}

Equations should be centered and numbered. They should be referenced in the text as, for
instance, Eq. (\ref{ena}). Please use adequate letter fonts (including size and type). A typical
equation is as follows
\begin{equation}\label{ena}
f=\sqrt{(\alpha-\beta)^{T}(\gamma-\delta)},
\end{equation}
where the variables are unknown.


\section{Figures}

Figures should be centered and referenced in the text as, for example, Figure~\ref{dva}. Please use adequate
letter fonts (including size and type) and line width. The example follows.

As you see in Figure~\ref{dva} below, the figure caption is
automatically generated in the proper format.
\begin{figure}
\center
\includegraphics[width=11.5cm]{ARKfigure}
\caption{Please write your figure caption here.}\label{dva}
\end{figure}


\section{Tables}

Tables should be centered and referenced in the text as, for
example, Table \ref{tri}. Here is an example of a typical table.

\begin{table}
\caption{Please write your table caption here.}
\label{tri}
\begin{tabular}{p{2cm}p{2.4cm}p{2cm}p{4.9cm}}
\hline\noalign{\smallskip}
Classes & Subclass & Length & Action Mechanism  \\
\noalign{\smallskip}\svhline\noalign{\smallskip}
Translation & mRNA$^a$  & 22 (19--25) & Translation repression, mRNA cleavage\\
Translation & mRNA cleavage & 21 & mRNA cleavage\\
Translation & mRNA  & 21--22 & mRNA cleavage\\
Translation & mRNA  & 24--26 & Histone and DNA Modification\\
\noalign{\smallskip}\hline\noalign{\smallskip}
\end{tabular}
\end{table}


\section{Referencing}

The reference list should be in an alphabetical order. The provided
class file will produce the format you see, which is correct. A citation
in the paper should have the form \cite{contrib}, \cite{mono},
\cite{journal}, \cite{online} for a contribution in conference
proceedings, monograph, journal article and on line document.


\section{Conclusions}
Your manuscript must be sent by e-mail to jadran.lenarcic@ijs.si.

\begin{acknowledgement}
The research work reported here was made possible by Grant No. 12345 of ARK Research Council.
\end{acknowledgement}

\begin{thebibliography}{99.}
% Use the following syntax and markup for your references

% Contribution in conference proceedings
\bibitem{contrib} Broy, M.: Software engineering -- from auxiliary to key technologies. In: Broy, M., Dener, E. (eds.) Software Pioneers, pp. 10-13. Springer, Heidelberg (2002)

% Monograph
\bibitem{mono} Geddes, K.O., Czapor, S.R., Labahn, G.: Algorithms for Computer Algebra. Kluwer, Boston (1992)

% Journal article
\bibitem{journal} Kamburger, C.: Quasimonotonicity, regularity and duality for nonlinear systems of partial differential equations. Ann. Mat. Pura. Appl. \textbf{169}, 321--354 (1995)

% Online Document
\bibitem{online} Qoody, J.: Effective substances. In: The Dictionary of Substances and Their Effects. Royal Society of Chemistry (1999) Available via DIALOG. \\
\url{http://www.rsc.org/dose/title of subordinate document. Cited 15 Jan 1999}

\end{thebibliography}

\end{document}

















